Author's response to reviews

Title: Determinants of rates of exclusive breast feeding in Nigeria

Authors:

Kingsley E Agho (k.agho@uws.edu.au)
Michael J Dibley (mdibley@health.usyd.edu.au)
Justice I Odiase (justiceodiase@yahoo.com)
Sunday M Ogbonmwan (ogbonmwasmaltra@yahoo.co.uk)

Version: 2 Date: 11 May 2010

Dear Editor-in-Chief

DETERMINANTS OF EXCLUSIVE BREASTFEEDING IN NIGERIA:

Agho KE, Dibley MJ, Odiase JI and Ogbonmwan SM

We would like to re-submit our manuscript entitled ‘Determinants of Non-exclusive breastfeeding in Nigeria’ with manuscript number: ID 1625006249323810.

The title of the paper has changed from Determinants of Non-exclusive breastfeeding in Nigeria to “Determinants of exclusive breastfeeding in Nigeria” as suggested by the reviewers. Answers to the reviewers’ comments have been addressed (see attached document “Agho, et al _BMC_Answer_to_Reviewers”).

Please address any correspondence to:

Dr. Kingsley Agho,
Research Fellow in Biostatistics,
School of Medicine,
University of Western Sydney,
Parramatta Campus,
Locked Bag 1797,
PENRITH NSW DC1797 Australia
Phone: +612 9685 9429
Fax: +612 9685 9554
Email: k.ago@uws.edu.au

Thank you for your cooperation.

Yours sincerely

Kingsley Agho